Author’s response to reviews

Title: Nurse-led video-coaching interventions in childhood, adolescent and young adult cancer survivors (REVIVER): a protocol for mixed methods feasibility research

Authors:

Eline Bouwman (eline.bouwman@radboudumc.nl)

Rosella Hermens (rosella.hermens@radboudumc.nl)

Nicole Blijlevens (nicole.blijlevens@radboudumc.nl)

Judith Prins (judith.prins@radboudumc.nl)

Jacqueline Loonen (jacqueline.loonen@radboudumc.nl)

Version: 2 Date: 15 Oct 2019

Author’s response to reviews:

E. Bouwman
Centre of Expertise for Cancer Survivorship
Radboud University Medical Centre, Department of Haematology
Geert Grooteplein Zuid 10 (route 506)
0031 (0)24 361 3318
6525 GA Nijmegen, the Netherlands
eline.bouwman@radboudumc.nl

Dr. Gillian Lancaster
Editor-in-Chief
Pilot and Feasibility Studies

October 15, 2019

Dear Dr. Lancaster,

We are very grateful for the opportunity to resubmit our manuscript and would like to thank you for your comments and suggestions in your kind letter “Your submission to Pilot and Feasibility Studies - PAFS-D-19-00119R1”. We have carefully adapted our revised manuscript according to these comments.

Please find enclosed 3 attachments:
1. Document “Responses to the reviewers’ comments (2)”

2. Our revised manuscript with visible track changes and comments: “TC_Revised manuscript PAFS-D-19-00119 (2)”

3. Our revised manuscript: “Revised manuscript PAFS-D-19-00119 (2)”

The document “Responses to the reviewers’ comments” contains point-by-point responses to the reviewers’ comments. Changes in the text are clearly visible, both in this document and in the manuscript by means of visible track changes.

We highly value your useful contribution, which we believe has led to considerable improvement of our manuscript. We hope that the editors and reviewers will be satisfied with the further amendments which we have made to the manuscript. We appreciate it very much that you are willing to reconsider our manuscript for publication after these revisions. If you need any further information, please do not hesitate to contact us.

Yours faithfully,

On behalf of all authors,

Eline Bouwman, MSc